**Author’s response to reviews**

**Title:** The epidemiology of infectious gastroenteritis related reactive arthritis in U.S. military personnel: a case-control study

**Authors:**

Jennifer A Curry (jennifer.curry@med.navy.mil)
Robert Gormley (Robert.Gormley@med.navy.mil)
Mark S Riddle (Mark.Riddle@med.navy.mil)
David R Tribble (dtribble@idcrp.org)
Chad K Porter (chad.porter@med.navy.mil)

**Version:** 3 **Date:** 3 September 2010

**Author’s response to reviews:**

As requested, the following formatting changes were made to the editorially approved manuscript:

1) Author list written in preferred ‘First name Initial Last name’ style.

2) Reference 48 removed from the bibliography (as it had been removed from the text), and the remaining citations were re-numbered. The missing information from citation 42 has been added.

3) Track changes were removed

4) Visible vertical lines were removed from the tables.